Author's response to reviews

Title: Use of complementary and alternative medicine by those with a chronic disease and the general population - results of a national population based survey

Authors:

Amy Metcalfe (amy.metcalfe@albertahealthservices.ca)
Jeanne Williams (jvawilli@ucalgary.ca)
Jane McChesney (jane.mcchesney@albertahealthservices.ca)
Scott B Patten (patten@ucalgary.ca)
Nathalie Jetté (nathalie.jette@albertahealthservices.ca)

Version: 7 Date: 6 October 2010

Author's response to reviews:

Minor typographical changes were made to the manuscript.